Author's response to reviews

Title: Near-highway exposure to motor vehicle pollutants: Emerging evidence of cardiac and pulmonary health risks

Authors:

Doug Brugge (dbrugge@aol.com)
John Durant (john.durant@tufts.edu)
Christine Rioux (Christinerioux@aol.com)

Version: 2 Date: 5 January 2007

Author's response to reviews:

This is reformatted to meet the journal's specifications.